Author's response to reviews

Title: Transcriptional profiles of pilocytic astrocytoma are related to their three different locations, but not to radiological tumor features

Authors:

Krzysztof Zakrzewski (krzysztof.zakrzewski_xl@wp.pl)
Michal Jarzab (mjarzab@io.gliwice.pl)
Małgorzata Oczko-Wojciechowska (gosiaoczko@io.gliwice.pl)
Aleksandra Pfeifer (apfeifer@io.gliwice.pl)
Barbara Jarzab (bjarzab@io.gliwice.pl)
Pawel P Liberski (pawel.liberski@umed.lodz.pl)
Magdalena Zakrzewska (magdalena.zakrzewska@umed.lodz.pl)

Version: 5
Date: 25 September 2015

Author's response to reviews: see over
Dear Editor,

Thank you very much for the comment. In the revised manuscript we included explicit information saying that the PCR-validation was performed on independent set of tumors. We also add such statement to the description of Figure 6. It is important for our team to present results in a correct manner. According to the Copy Edit recommendation the text was corrected and the style of written English was improved during linguistic correction.

I hope that the manuscript will be of interest for readers of BMC Cancer.

Yours faithfully,

Magdalena Zakrzewska

Medical University of Lodz
Al. Kościuszki st. 4
PL 90-419 Lodz, Poland